Author's response to reviews

Title: Cultural Competence among maternal health care providers in Bahir dar city Adminstration, Northwest Ethiopia: Cross Sectional Study

Authors:

Amanu A Emiru (aragawamanu@yahoo.com)
Tegbar Y Sendekie (tyigzaw@jhpiego.net)
Desalegn T Medhanye (meladesu@gmail.com)
Wubalem G Bayleyegne (Wubie2001@gmail.com)

Version:2 Date: 14 August 2014

Author's response to reviews:

based on your requirement for revision, we included line numbering and page number for the manuscript. Also, tables are put after the reference section.